Author's response to reviews

Title: British pain clinic practitioners' recognition and use of the bio-psychosocial pain management model for patients when physical interventions are ineffective or inappropriate: results of a qualitative study

Authors:

Geoffrey Harding (geoffrey.harding@pms.ac.uk)
John Campbell (John.campbell@pms.ac.uk)
Suzanne Parsons (suzanne.parsons@PickerEurope.ac.uk)
Anisur Rahman (anisur.rahman@ucl.ac.uk)
Martin Underwood (m.underwood@warwick.ac.uk)

Version: 7 Date: 15 March 2010

Author's response to reviews:

The author list has been altered to correspond with the submission system.

The abstract has been reformatted to use the requested headings Background, Methods, Results, Conclusions